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DAILY CONSTRUCTION PROGRESS REPORT 

-- INSPECTOR -- 
 
Project: White Oak Mine Reclamation        Date: 10/26/2003    M   T   W   Th   F   S   Su 
 
Crew:   Two                             Supervisor:   Wayne Helsel         Hours: 9:00 am             to  4:00 pm        
 
Equipment Used:  1- Track Hoe, 2- Dozers                                                                                                    
 
 
General description of work performed, equipment/material deliveries, etc: 
 
When I first arrived, a track hoe was moving spoil material away from the edge and piling it in Pit E at the 
mine site.  A dozer was pushing spoil into Pit D from the southeast.                                                               
 
A surveyor from Whiskey Creek Mining Services surveyed the Toe Line for Pit E and the upper pond.        
 
Later during the day, the dozer that was earlier pushing fill in Pit D and another dozer began pushing the 
spoil in front of Pit E back to the highwall of Pit E.                                                                                           
 
Spoil had been pushed to the faces of the portals, no spoil had been pushed into the portals.  Wayne 
stated that they are waiting the arrival of MSHA personnel to observe backfilling the portals.  He thought it 
would take place on Monday the 27th.                                                                                                             
 
Roads are extremely dusty.                                                                                                                             
 
No salvage work was taking place today.  Yesterday, the salvage crew had cut through the stacking tube 
and tried to pull it down, but couldn’t do it.  They plan to try again, probably tomorrow.                                  
 
 
YARDS MOVED:     0  CY 
TOTAL MOVED TO DATE     0  CY 
TOPSOIL MOVED:     0  CY 
AREA SEEDED:     0   
RIPRAP         0   
 
 
POTENTIAL ISSUES:                                                                                                                                      
 
 
Visitors & purpose:    None                                                                                                                              
 
 
PICTURES TAKEN       YES        NO 
 
 
Temp: 50’s to 70’s  Comments: 
Sky: Clear     
Ground: Dry      
 
DOGM:    David W. Darby                                                               10/26/2003    
 


